Author's response to reviews

Title: Prescribing trends of antipsychotics in youth receiving income assistance: results from a retrospective population database study

Authors:
Andrea L Murphy (andrea.murphy@dal.ca)
David M Gardner (david.gardner@dal.ca)
Charmaine Cooke (Charmaine.Cooke@gov.ns.ca)
Steve Kisely (s.kisely@uq.edu.au)
Jean Hughes (jean.hughes@dal.ca)
Stan P Kutcher (Stanley.kutcher@iwk.nshealth.ca)

Version: 3 Date: 22 May 2013

Author's response to reviews:

Dear BMC Psychiatry editorial team:

We have substantively revised the abstract and background of the paper to improve readability. Throughout of the remainder of the paper we have made minor corrections in syntax and grammar.

Sincerely,

Andrea Murphy, on behalf of all authors